**Author’s response to reviews**

**Title:** Multifactorial intervention for hip and pelvic fracture patients with mild to moderate cognitive impairment: Study protocol of a dual-centre randomised controlled trial (OF-CARE)

**Authors:**

Anja Dautel (anja.dautel@rbk.de; adautel@web.de)

Tobias Eckert (tobias.eckert@bethanien-heidelberg.de)

Michaela Gross (michaela.gross@rbk.de)

Klaus Hauer (khauer@bethanien-heidelberg.de)

Martina Schäufele (m.schaeufele@hs-mannheim.de)

André Lacroix (andre.lacroix@bethanien-heidelberg.de)

Ingrid Hendlmeier (i.hendlmeier@hs-mannheim.de)

Bastian Abel (bastian.abel@bethanien-heidelberg.de)

Rebekka Pomiersky (rebekka.pomiersky@bethanien-heidelberg.de)

Julia Gugenhan (julia.gugenhan@rbk.de)

Gisela Büchele (gisela.buechele@uni-ulm.de)

Katrin C. Reber (k.reber@uke.de)

Clemens Becker (clemens.becker@rbk.de)

Klaus Pfeiffer (klaus.pfeiffer@rbk.de)

**Version:** 1  **Date:** 13 Mar 2018

**Author’s response to reviews:**

Dear Reviewer,

Please find the required corrections below in the order you listed the items in your last letter.

I want to emphasize that the failures which occurred during submission process are solely in my responsibility and not in the responsibility of the co-authors.
All authors approved the submitted revised manuscript and agreed to authorship and order of authorship for this revised manuscript.

---------------------
1. Email addresses

The email addresses for all authors are included on the title page and it is ensured that these email addresses match the email addresses provided in the editorial manager system. The corresponding author is indicated.

2. Consent for publication

"Not applicable" is stated in this section.

3. Consent to participate

A statement on consent to participate in the “Ethics approval and consent to participate” section of the Declarations is included.

4. SPIRIT Guidelines

A completed SPIRIT checklist is submitted as additional file.

5. Abstract

The box surrounding the abstract is removed.

6. Study protocol specific requests

a) Copies of all ethical approval and funding approval in German and English are sent via email.

b) Funding

The study is funded by the German Federal Ministry of Education and Research.
It is stated in the manuscript that there are no competing interests.

It is confirmed that our study protocol has not undergone peer-review by the funding body.

c) Study status

It is confirmed that the study is ongoing (recruitment of participants is completed at both study sites, intervention as well as follow-up assessments will run until the end of 2018).

d) Related Articles

It is stated that no publications have been published containing the results of the study being described in the study protocol.

7. Clean copy

The manuscript does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Thank you very much for processing our submission.

Kind Regards,